Author’s response to reviews

Title: Delayed-onset of progressive pseudorheumatoid dysplasia in a Chinese adult with a novel compound WISP3 mutation: a case report

Authors:

Qiong Hu (huqiongyi131@163.com)
Jing Liu (14210700034@fudan.edu.cn)
Yi Wang (yawing@fudan.edu.cn)
Jiu Wang (theresajcwang@gmail.com)
Hui Shi (shihui_sjtu@sina.com)
Yue Sun (winniesun2015@163.com)
Xin Wu (ltzx2004308@163.com)
Jialin Teng (tengteng8151@sina.com)
yangchengde@sina.com Yang (yangchengde@sina.com)

Version: 2 Date: 23 Nov 2017

Author’s response to reviews:

MGTC-D-17-00211R1

Delayed-onset of progressive pseudorheumatoid dysplasia in a Chinese adult with a novel compound WISP3 mutation: a case report

Qiongyi Hu; Jing Liu; Yi Wang; Jiucun Wang; Hui Shi; Yue Sun; Xinyao Wu; Jialin Teng; Chengde Yang, Ph.D

Dear Editor Menard,

Thank you very much for your comments that will be of great help to improve our manuscript. We have made revision carefully for this manuscript according to your comments. The point-by-point response to your comments is listed as follows:
1. Please provide the email addresses of all authors on the title page.

Answer: The email addresses of all authors are listed on the title page.

2. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed. e.g. "conform" in the Consent for publication section instead of "confirm".

Answer: Thank you very much for your constructive comments to improve our paper. We have previously improved our manuscript using the Nature Research Editing Service. We checked throughout the paper and made corrections where needed.

3. Please clarify in the Consent for publication section what the patients consented to be published, e.g. their medical data, images, etc.

Answer: We have clarified in the Consent for publication section that the patients consented to be published their medical data and images (as showed in the Consent for publication section, line 3, page 9).

4. As ethical approval is not usually acquired for case reports, please can you confirm whether the patients in this case report received standard care. You do not need to remove the ethics information, please leave this in the Ethics approval and consent to participate section.

Answer: The patients in this case report received standard care, including physical examination, radiography examination and blood collecting for genetic analysis.

5. Please amend the corresponding author's name in Editorial Manager, currently their email address has been put in instead of their first name.

Answer: We have amended the corresponding author's name in Editorial Manager. The first name is Chengde.
6. Please remove the second title page which is located just before the tables, as this is not needed. Please simply upload the tables.

Answer: We have removed the second title page and upload the simple table.

7. Thank you for including the CARE checklist, as this is no longer required at this stage, please remove the checklist and include a statement in the Case presentation section stating that the CARE guidelines were followed in reporting this case.

Answer: We have stated in the Case presentation section that CARE guidelines were followed (as showed in the Case presentation section, line 8, page 5).

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: I'm very appreciated for your recommendations. We have uploaded our manuscript as a clean version. And the table and figures are also clean versions.

We trust that we have adequately addressed your concerns and our revision should be sufficient to merit acceptance of our manuscript for publication in BMC Medical Genetics. We deeply appreciate your consideration of the paper for publication in your journal, and shall look forward to hearing from you at your earliest convenience.

Sincerely yours,

Chengde Yang

Chengde Yang, Ph.D
Department of Rheumatology and Immunology,
Ruijin Hospital,
Shanghai Jiao Tong University School of Medicine,
No. 197 Ruijin Second Road, Shanghai 200025, China.
Tel: (86)-21-64370045
Fax: (86)-21-54109718
E-mail: yangchengde@sina.com;